Author's response to reviews

Title: Mechanisms of tumor-promoting activities of nicotine in lung cancer

Authors:

Alex I Chernyavsky (sgrando@uci.edu)
Igor B Shchepotin (sgrando@uci.edu)
Valentin Galitovkiy (sgrando@uci.edu)
Sergei A Grando (sgrando@uci.edu)

Version: 4  Date: 26 January 2015

Author's response to reviews: see over
Dear Editor,

I am glad to submit for your consideration the revised version of our manuscript titled "MECHANISMS OF TUMOR-PROMOTING ACTIVITIES OF NICOTINE IN LUNG CANCER."

As requested by Reviewer 1, the title has been changed to reflect the focus on lung cancer, and the minor changes in the figure legends were introduced. All changes are highlighted in the text.

Reviewer 2 had no suggestions.

I hope the revised manuscript can be accepted for publication.

Thank you

On behalf of the authors.

Sincerely,

Sergei A. Grando, MD, PhD, DSc
Professor,
Departments of Dermatology and Biological Chemistry
Cancer Center & Research Institute